

Flasche, S; Le Polain de Waroux, O; OBrien, KL; Edmunds, WJ (2015) The Serotype Distribution among Healthy Carriers before Vaccination Is Essential for Predicting the Impact of Pneumococcal Conjugate Vaccine on Invasive Disease. PLoS computational biology, 11 (4). e1004173. ISSN 1553-734X DOI: 10.1371/journal.pcbi.1004173

Downloaded from: http://researchonline.lshtm.ac.uk/2159817/

DOI: 10.1371/journal.pcbi.1004173

Usage Guidelines

 $Please \ refer \ to \ usage \ guidelines \ at \ http://researchonline.lshtm.ac.uk/policies.html \ or \ alternatively \ contact \ researchonline@lshtm.ac.uk.$ 

Available under license: http://creativecommons.org/licenses/by/2.5/

**RESEARCH ARTICLE** 

# The Serotype Distribution among Healthy Carriers before Vaccination Is Essential for Predicting the Impact of Pneumococcal Conjugate Vaccine on Invasive Disease

Stefan Flasche<sup>1</sup>\*, Olivier Le Polain de Waroux<sup>1</sup>, Katherine L. O'Brien<sup>2</sup>, W. John Edmunds<sup>1</sup>

1 London School of Hygiene and Tropical Medicine, London, United Kingdom, 2 International Vaccine Access Center, Johns Hopkins Bloomberg School of Public Health, Baltimore, Maryland, United States of America

\* Stefan.Flasche@lshtm.ac.uk

## Abstract

Pneumococcal conjugate vaccines (PCVs) have substantially reduced morbidity and mortality of pneumococcal disease. The impact of the 7-valent PCV on all-serotype invasive pneumococcal disease (IPD) among children was reported to vary between high-income countries. We investigate the ability to predict this heterogeneity from pre-vaccination data. We propose a parsimonious model that predicts the impact of PCVs from the odds of vaccine serotype (VT) among carriers and IPD cases in the pre-PCV period, assuming that VT are eliminated in a mature PCV programme, that full serotype replacement occurs in carriage and that invasiveness of the NVT group is unchanged. We test model performance against the reported impact of PCV7 on childhood IPD in high-income countries from a recent meta-analysis. The odds of pre-PCV7 VT IPD, PCV schedule, PCV coverage and whether a catch up campaign was used for introduction was gathered from the same analysis. We conducted a literature review and meta-analysis to obtain the odds of pre-PCV7 VT carriage in the respective settings. The model predicted the reported impact on childhood IPD of mature PCV programmes; the ratio of predicted and observed incidence risk ratios was close to 1 in all settings. In the high income settings studied differences in schedule, coverage, and catch up campaigns were not associated with the observed heterogeneity in impact of PCV7 on childhood all-serotype IPD. The pre-PCV7 proportion of VT IPD alone also had limited predictive value. The pre-PCV7 proportion of VT carriage and IPD are the main determinants for the impact of PCV7 on childhood IPD and can be combined in a simple model to provide predictions of the vaccine preventable burden of IPD.

## Author Summary

Pneumococcal vaccines (PCVs) that protect children against 7, 10 and 13 of the most pathogenic pneumococcal serotypes have substantially reduced childhood morbidity and



## GOPEN ACCESS

**Citation:** Flasche S, Le Polain de Waroux O, O'Brien KL, Edmunds WJ (2015) The Serotype Distribution among Healthy Carriers before Vaccination Is Essential for Predicting the Impact of Pneumococcal Conjugate Vaccine on Invasive Disease. PLoS Comput Biol 11(4): e1004173. doi:10.1371/journal. pcbi.1004173

**Editor:** Daniel Martin Weinberger, Yale School of Public Health, UNITED STATES

Received: October 13, 2014

Accepted: February 5, 2015

Published: April 16, 2015

**Copyright:** © 2015 Flasche et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

**Funding:** The authors received no specific funding for this work

**Competing Interests:** The authors have declared that no competing interests exist.

mortality. A recent analysis that evaluated the impact of the 7 valent PCV in multiple high income settings in North America, Europe and Oceania found that the magnitude of all-serotype invasive pneumococcal disease reduction varied greatly between settings (24%-83%). We explored potential sources for that variation, including differences in disease epidemiology before vaccination, vaccine coverage, vaccine schedules and the use of catch-up campaigns for introduction. We find that differences in reported disease impact among mature PCV programmes are likely to be unrelated to the differences in the vaccine programme but can be predicted from a simple model based on pre-vaccination epidemiology, in particular the proportion of vaccine serotypes detected among patients with invasive pneumococcal disease and the proportion of vaccine serotypes that are found in the nasopharynx of healthy individuals. This model presents a useful tool to estimate the potential impact of PCVs (as a relative rate reduction), highlights the essential role of pre-vaccination carriage in healthy individuals for disease impact of PCVs and can estimate the prevented burden of disease where disease surveillance is unavailable.

## Introduction

The Word Health Organisation estimates that *Streptococcus pneumoniae* is associated with about 5% of all-cause child mortality globally; over 90% of these pneumococcal deaths occur in low income countries [1]. Pneumococcal conjugate vaccines (PCVs) are part of the routine infant immunization schedule in most high income countries, resulting in a substantially reduced burden of serious pneumococcal disease [2–4]. PCVs are also being introduced into the routine vaccination programmes of low and middle income countries, partly with the financial support of Gavi, the Vaccine Alliance [5–7]. PCVs provide protection against nasopharyngeal carriage and disease for serotypes included in the vaccine (VT); these serotypes have been associated with the majority of invasive pneumococcal disease (IPD) globally [8]. Protection against VT nasopharyngeal carriage opens an ecological niche which is filled by the non-vaccine pneumococcal serotypes (NVT); a process termed serotype replacement [9–11]. This increase in NVT colonization prevalence results in an increased rate of NVT disease; however, because these serotypes are inherently less likely to cause disease among young children than VT strains, there is a substantial net benefit [12]. Understanding the interplay between VT protection and NVT replacement is essential for the assessment of the total impact of PCVs [13,14].

Despite being consistently beneficial, substantial heterogeneity in the relative impact of PCV7 on all-serotype IPD rates has been observed across settings, with impact estimates in children younger than 5 years ranging from 24% to 83% in mature programmes [12]. This heterogeneity is thought to result from interactions of vaccine coverage, vaccination schedule, serotype distribution, demographic structure and social mixing patterns, catch up campaigns at introduction, time since PCV introduction, and disease surveillance sensitivity. The contribution of each of those factors to the observed heterogeneity in PCV impact on all-serotype IPD is unclear. PCVs are amongst the most expensive vaccines that are routinely used for infant vaccination. Although the Advance Market Commitment (AMC) and the support of Gavi, the Vaccine Alliance substantially reduced the PCV price for low income countries [15] and pooled procurement might help reducing the costs for middle income countries [16] PCVs pose a considerable investment that requires robust evidence about its likely impact. A better understanding of the main factors that determine the impact of PCVs in PCV-naive countries, as well as for the assessment of the likely impact of future PCV compositions and to inform programme

maintenance justifications. While better impact predictions may help with a faster introduction of PCVs globally, the justification of existing pneumococcal immunisation programmes will become particularly important for countries that have introduced PCVs with financial support from Gavi, the Vaccine Alliance and will graduate from that support. These countries have to evaluate the merits of vaccination at or below the agreed tail price of PCV under the AMC agreements, however, measuring disease impact is only possible in a limited number of countries. Disease impact models are therefore important for many countries.

Available methods to predict the likely impact of PCV on disease include models accounting for carriage and disease serotype distribution and replacement as the main drivers for PCV impact [14,17,18] and more complex transmission models [19–21]. To date little validation of the models capability to accurately predict post vaccination changes in pneumococcal disease is available. We evaluate the ability of the pre-vaccination pneumococcal serotype distribution in both nasopharyngeal carriage and disease, vaccine coverage, schedule and catch up campaigns to predict the impact of PCV7 on invasive pneumococcal disease in children less than five years old and the importance of each of those factors for the accuracy of the prediction.

## **Materials and Methods**

## A predictor of PCV impact on pneumococcal disease

To predict the impact of PCV on pneumococcal disease we employ a model that is similar to previous approaches and uses changes in pneumococcal carriage to predict the impact of PCV on IPD [14,17,18,22].

For simplicity we assume a perfectly monitored homogenous population. Note that the methods can be derived similarly if imperfect sensitivity of carriage and/or disease surveillance and a heterogeneous population is assumed, as long as surveillance sensitivity and population heterogeneity does not change after vaccination. In this population and in the absence of PCV vaccination the rate of pneumococcal disease (D) per person-time can be expressed as a function of the rate of carriers (C) per person-time and the average ratio at which a carriage episode results in disease ( $\Phi = D/C$ ), stratified by vaccine and non-vaccine serotypes respectively:

$$D = C_{vt} \Phi_{vt} + C_{nvt} \Phi_{nvt}.$$

The parameter  $\Phi$  is also called the case to carrier or invasiveness ratio for VT or NVT, which corresponds to the mean of the serotype specific invasiveness ratios weighted by serotype-specific carriage prevalence. We further assume that (i) vaccine serotype carriage will eventually be eliminated through routine use of PCV ( $C_{vt}^* = 0$ ), the superscript star indicating post vaccination, (ii) that a proportion,  $\lambda \in [0, 1]$  of pre-PCV VT carriage is replaced by NVT carriage ( $C_{nvt}^* = \lambda C_{vt} + C_{nvt}$ ), and (iii) that the invasiveness ratio of the non-vaccine serotypes group remains unchanged after vaccination ( $\Phi_{nvt}^* = \Phi_{nvt}$ ). Then the rate of pneumococcal disease in a mature PCV-vaccination programme ( $D^*$ ), when the programme has been in place long enough for direct and indirect effects to become fully established, can be expressed as a function of pre-vaccination pneumococcal carriage rates and the invasiveness of the NVT group of serotypes:

$$D^* = (\lambda C_{vt} + C_{nvt})\Phi_{nvt}.$$

Then the expected Incidence Rate Ratio (IRR), simplifies to:

$$IRR = \frac{D^*}{D} = \frac{\lambda c + 1}{d + 1}$$

where *c* and *d* are the odds of VT carriage and disease, respectively prior to vaccination. That is

 $c = C_{vt} / C_{nvt}$  and  $d = D_{vt} / D_{nvt}$ . Note that, although technically *c* and *d* are the odds based on disease and carriage rates, it is equivalent to calculate the odds based on counts even if carriage and disease data arise from samples of different sizes or proportions of VT and NVT among pneumococcal carriage and disease rates. For convenience we will mainly refer to proportions hereafter.

Therefore, if *c* and *d* can be obtained from representative samples of the population, the expected percentage change in IPD after vaccination, one of the key measures of vaccine impact, can be estimated from pre-vaccination data alone (given an informed assumption on the level of replacement ( $\lambda$ ) is available).

## Data for the validation of the prediction model

To test the performance of our predictor and the potential importance of other factors in accounting for the observed heterogeneity of PCV impact we compared the change in IPD rate after routine use of PCV7 with model predictions that use pre-vaccination data on the proportion of VT in IPD and pneumococcal carriage in a sample of the population from the same study site or country. We assumed that carriage and disease in these sub-population were representative of that in the respective study site or country. We further assumed that the mean duration of carriage for VT and NVT is similar which allowed the use of carriage prevalence for the calculation of the odds of VT carriage. We studied two predictions: (i) our main prediction that assumes complete serotype replacement in nasopharyngeal carriage ( $\lambda = 1$ ) as is observed in most settings where PCV7 was introduced for routine vaccination and (ii) an alternative prediction that assumes no serotype replacement in carriage ( $\lambda = 0$ ). This prediction requires no carriage data because it reduces to  $IRR = \frac{D_{mt}}{D}$ . This prediction illustrates the impact of assuming that all vaccine preventable IPD is eliminated and does not take into account serotype replacement

**Observed IRR.** The impact of 7-valent pneumococcal conjugate vaccination on invasive pneumococcal disease, as measured by incidence rate ratios (IRR), has recently been summarized from country specific IPD disease rate surveillance data pre- and post- PCV routine use [12]. We used the IRR estimates on all serotype IPD in children under 5 years of age three years after the start of routine vaccination as the observed impact against which we assessed the predicted impact and the role of potential modifying factors on those estimates (see Table 1). Three years has been deemed sufficient time to allow for almost complete elimination of VT disease and completion of serotype replacement in carriage and disease in under 5 year old children in the UK and US [4,23]. This may vary in other countries, depending on factors including vaccine coverage and the use of catch up campaigns. However, Feikin and colleagues [12] reported a reduction of VT IPD in under 5 year old children that was significantly lower than 90% in the 3 years after introduction of vaccination in only 1 out of 14 settings. The mean reduction after 3 years was 91%. Sensitivity analyses which compare model predictions against observed impact after 4, 5 and 6 years are reported in the appendix. Only data in under 5 year old children were considered because carriage information in older age groups is sparse and indirect effects, both VT elimination and serotype replacement, in those age groups, take longer to reach maximum impact. To limit the potential bias of surveillance artefacts we excluded all settings that reported a decrease of NVT IPD incidence after the introduction of PCV. Hence Norway was excluded from the main analysis but was included in our assessment of model sensitivity and reported in the appendix.

**Pre-vaccination IPD serotype distribution data.** In addition to the observed impact of PCV7 on IPD, Feikin et al. [12] also reported the pre vaccination proportion of VT IPD for each surveillance setting (see <u>Table 1</u>). The proportion of NVT IPD was calculated as 1-proportion of VT IPD.

	Vaccine schedule	Catch up	PCV7 coverage*	Sites	VT %	N IPD	Sites	VT%	N PNC	Source	Observed	Predicted $(\lambda = 1)$	Predicted / Observed
				Invasive Pneumococcal Disease			Pneumo	arriag		IRR			
ABCs	3+1	Y	39%	Active Bacterial Core surveillance USA	88%	715	Atlanta	61.5%	91	[45]	0.33 (0.27 to 0.41)	0.31 (0.23 to 0.44)	0.95 (0.64 to 1.41)
AIP	3+1	Y	46%	Alaska	81%	95	8 rural villages	55.4%	377	<u>46</u>			
							Anchorage	53.8%	171	[47]			
							Pooled estimate	54% (25% to 77%)			0.40 (0.22 to 0.73)	0.41 (0.21 to 0.87)	1.01 (0.42 to 2.68)
AUSI	3+PPV	Y	77%	Australia Indigenous	66%	100	Remote community in northern Australia	30.0%	80	[ <u>48]</u>	0.60 (0.29 to 1.22)	0.49 (0.35 to 0.65)	0.80 (0.37 to 1.72)
AUSN	3+0	Y	90%	Australia non- Indigenous	89%	1245	Darwin	80.8%	125	[ <u>48]</u>	0.46 (38 to 0.57)	0.57 (0.40 to 0.90)	1.25 (0.81 to 2.05)
DEN	2+1	Y	89%	Denmark	70%	455	Roskilde	33.2%	247	[ <u>49]</u>	0.39 (0.27 to 0.57)	0.45 (0.38 to 0.53)	1.15 (0.77 to 1.73)
E&W	2+1	Y	89%	England & Wales	78%	3450	Hertfordshire	67.1%	85	[9]			
							Oxford	56.4%	349	[ <u>50]</u>			
							London	57.1%	119	[51]			
							Sheffield	68.5%	111	<u>52</u>			
							Oxfordshire	56.7%	60	<u>53</u>			
							Oxfordshire	64.2%	95	[54]			
							Pooled estimate	61% (54% to 68%			0.55 (0.50 to 0.61)	0.57 (0.48 to 0.69)	1.03 (0.85 to 1.30)
NAV	3+1	Y	68%	Navajo and White Mountain Apaches	58%	100	Navajo and White Mountain Apaches	38.0%	258	[ <u>55]</u>	0.76 (0.38 to 1.28)	0.68 (0.52 to 0.85)	0.89 (0.42 to 1.86)
NCK	3+1	Y	47%	Northern California Kaiser Permanente	81%	88	California	60.0%	5	[ <u>56]</u>	0.70 (0.38 to 1.53)	0.48 (0.19 to 204.55)	0.71 (0.22 to 2.45)
NLD	3+1	Ν	94%	Netherlands	74%	245	Western Netherlands	55.4%	213	[57]			
							Rotterdam	54.5%	220	[ <u>58]</u>			
							Pooled estimate	52% (21% to 79%)			0.53 (0.33 to 0.86)	0.56 (0.33 to 1.17)	1.06 (0.53 to 2.45)

#### Table 1. Description of the data.

Description of study sites contributing information about pre-PCV IPD cases [12] and NP colonization characteristics, in children less than 5 years of age. Where multiple studies on nasopharyngeal carriage per IPD setting were included also a pooled estimate (S1 Fig) is presented. \*average vaccine coverage in the first 3 years after introduction of PCV7

doi:10.1371/journal.pcbi.1004173.t001

**Pre-vaccination carriage serotype distribution data.** Studies with carriage prevalence estimates for those settings where the impact of PCV7 on IPD was reported, or any subset of that population, were identified through a systematic review, based on the citations retrieved in a recent systematic review of nasopharyngeal carriage in adults and children [24] (Table 1 and S1 Fig). In brief: MEDLINE and Embase electronic databases were used to retrieve articles up

to 23rd August 2013 (i.e. week 35) employing the following combination of search terms: '(pneumonia OR pneumoniae OR pneumococcal OR pneumococcus) AND (carriage OR colonization OR colonisation)' in the title or the keywords or the abstract. Articles which fulfilled the eligibility criteria by providing (i) pneumococcal nasopharyngeal carriage prevalence estimates (ii) in a population not previously exposed to PCV, with (iii) nasopharyngeal sampling and transport procedures as well as *S. pneumoniae* culture based on WHO guidelines and (iv) where the study was not restricted to specific serotypes or to *S.pneumoniae* with specific patterns of antibiotic sensitivity were retrieved.

Of the 376 studies that provided pre-PCV NP carriage estimates, 360 were excluded because they were not in populations where the disease impact was monitored, or because they did not report the proportion of under 5 year old children with nasopharyngeal carriage of any of the serotypes included in PCV7. 16 studies provided data on carriage serotype distribution before introduction of PCVs (<u>Table 1</u>). The carriage data were extracted by two reviewers independently and discrepancies were resolved by consensus.

We abstracted information on the number of carriers of encapsulated pneumococci (assuming that vaccination has no impact on non-encapsulated pneumococci and that those do not substantially contribute to serotype replacement) and the proportion carrying serotypes included in PCV7. In longitudinal studies with repeated sampling of the same individuals, the average VT and NVT carriage prevalence was calculated and the study size was assumed to be the number of unique study participants. We did not account for differences in age distribution due to non-random sampling from the under 5 year old population. Each of the included carriage studies was conducted only on a small subset of the respective populations monitored for IPD. We assumed that the carriage serotype distributions in those studies are representative of the carriage serotype distributions found in the populations that were monitored for IPD.

Where multiple studies on nasopharyngeal carriage were conducted within different subsets of the same population that were monitored for IPD, the results from those studies were combined via a Bayesian random effects meta-analysis.

**Data on other potential sources of heterogeneity in PCV impact.** Feikin et al. [12] reported information on potential sources of heterogeneity for the impact of vaccination including the average vaccine coverage in the first 3 years after introduction of PCV7, the routine vaccination schedule and whether PCV was introduced with a catch up campaign among older children (Table 1).

### Statistical analysis

To calculate the predicted IRRs and its corresponding distributions we assumed that both the proportion of VT among carriers and among IPD were samples from binomial distributions and drew 10,000 bootstrap samples. Where the proportion of VT carriers was derived through the Bayesian meta-analysis we drew the bootstrap samples from the respective posterior distribution instead (S2 Fig). Similarly, we assumed that the observed IRR were samples from log-normal distributions with confidence bounds matching those reported by Feikin et al [12].

We calculated the marginal distribution of observed IRRs for a specific schedule, coverage range or the use of a catch-up campaign upon implementation by bootstrap sampling from the respective observed IRR distributions (<u>S3</u> and <u>S4</u> Figs). The different impact of, e.g. a 3+1 schedule versus a 2+1 schedule, on the observed IRR was estimated through the ratio of the marginal distribution of IRRs of the 3+1 schedule settings and the marginal distribution of IRRs of the 2+1 schedule settings. A ratio centered around 1 indicates that the average impact of PCV7 in settings with either a 2+1 or a 3+1 schedule was similar.

We estimated the setting specific performance of our predicted IRRs by calculating the ratio of the predicted IRR to the observed IRR. A ratio of 1 is indicative of a perfect prediction. Non-parametric bootstrapping methods were used to infer confidence intervals.

All analysis was performed in R version 3.1 [25].

#### Results

The summary analysis of PCV impact on IPD three years after introduction of PCV7 [12] included data from 13 sites that met the inclusion criteria: indigenous and non-indigenous Australia, Calgary, Denmark, England and Wales, Crete, the Netherlands, Scotland, Switzerland, the US general population (Active Bacterial Core Surveillance), Alaska, Navajo Nation and Northern California (Kaiser Permanente). We identified childhood NP carriage information stratified by VT and NVT from the pre-PCV era from 9 of the 13 sites (Table 1). No carriage information was available in any healthy subpopulation of Calgary, Crete, Scotland and Switzerland. In England and Wales, Netherlands and Alaska more than one carriage study was identified. There was little or moderate heterogeneity between the study estimates of VT carriage proportion within the different sub populations of the same setting. Respective studies were pooled through Bayesian random effects meta-analysis to provide a single estimate of the proportion of VT among carriers for each setting (Table 1 and S2 Fig).

Fig. 1 illustrates how the serotype distributions in pneumococcal carriage and disease shape the predicted incidence risk ratios in the prediction model: the higher the proportion of VT in disease, the higher the predicted impact (lower IRR) and the lower the proportion of VT in carriage the higher the predicted impact. In particular this shows how serotype replacement in nasopharyngeal carriage and differences in serotype distribution in carriage prior to vaccine introduction can lead to vastly different vaccine impact predictions in two settings with similar contribution of VT to the pneumococcal disease burden.



**Fig 1.** Overview of the impact of model parameterisation on the predictions. The predicted IRRs in pneumococcal disease are shown in dependence of the proportions of VTs in carriage and IPD before vaccination. Shades of green represent predicted IRRs corresponding to a predicted reduction in disease. The height of the red triangle is determined by the extent of serotype replacement (shown for  $\lambda = 1$ ). The data on pre-vaccination PCV7 VT proportions in carriage and disease from the setting listed in Table 1 and their respective binomial confidence bounds are superimposed to illustrate the effect of differences in serotype distribution in both carriage and disease on the predicted impact of vaccination on pneumococcal disease.

doi:10.1371/journal.pcbi.1004173.g001

There was little difference between the average observed impact of PCV7 between settings with either different schedules or different coverage levels (<u>S3</u> and <u>S4</u> Figs). The impact in settings using a 2+1 schedule was slightly higher than in 3+1 schedule settings and the impact in 3+0 and 3+1 schedule settings was similar; ratio of IRRs 0.87 (0.20 to 2.91) and 1.00 (0.24 to 3.61). In settings with an average vaccine coverage of under 70% and those with a coverage between 70% to 90% the impact of PCV7 was similar to the impact in settings where vaccine coverage had been over 90%; ratio of IRRs 1.00 (0.39 to 2.42) and 0.95 (0.26 to 4.18). Settings which used a catch-up campaign for introduction of PCV7 reported on average a 25% (-65% to 325%) higher impact than those which had not.

Assuming no serotype replacement in carriage ( $\lambda = 0$ ) for the prediction led to consistent overestimation of vaccine impact (Fig. 2). With the assumption of complete serotype replacement in carriage ( $\lambda = 1$ ), however, we were able to closely predict the impact of routine PCV7 use on paediatric IPD 3 years after introduction (Fig. 2 and Fig. 3). The corresponding ratios of predicted and observed IRRs are provided in Table 1.

## Discussion

Routine pneumococcal conjugate vaccination has led to sustained reductions of all-serotype invasive pneumococcal disease in children, albeit of varying absolute and relative magnitudes across various countries and surveillance sites. Here we aimed to understand the factors related to the observed heterogeneity of PCV7 impact on IPD in children under five years of age 3 years after the start of routine vaccination so that estimates of that impact could reliably be produced for countries without disease impact data. Our analysis shows that the proportion of VT-IPD in the pre-PCV period, which is sometimes used as a measure of the potential vaccine preventable burden of *S. pneumoniae*, is consistently overestimating the observed impact of PCVs on overall IPD because it ignores the effect of serotype replacement. However, when supplemented by the odds of VT carriage pre-PCV in the same population our proposed model, assuming full serotype replacement in carriage, is highly predictive for the observed relative impact (IRR) of PCVs on overall IPD.



Fig 2. Model predictions in comparison to observed IRR. Comparison of predicted and observed impact of PCV7 on all-serotype IPD in children younger than 5 years, including confidence intervals, assuming no serotype replacement (left panel) or full serotype replacement (right panel) of VT carriage with NVT carriage. ABCs, AIP, AUSI, AUSN, DEN, E&W, NAV, NCK and NLD represent the Active Bacterial Core surveillance USA, Alaska (Calgary), Australia Indigenous, Australia non- Indigenous, Denmark, England & Wales, Navajo and White Mountain Apaches, Northern California Kaiser Permanente and Netherlands respectively.

doi:10.1371/journal.pcbi.1004173.g002





Fig 3. Model validation. Distributions of the ratio of predicted and observed IRR for each setting assuming full serotype replacement. A value of 1 represents perfect agreement between prediction and observation.

doi:10.1371/journal.pcbi.1004173.g003

We find that in the studied sites neither differences in schedule nor vaccine coverage substantially contributed to the heterogeneity of IPD impact observed from surveillance three years after the introduction of PCV7. While there is little doubt that sufficient vaccine coverage is essential for the success of a PCV programme we find that even in those settings where the average vaccine coverage during the first 3 years after implementation was below 70% the observed impact of vaccination was similar to settings with higher coverage. This may be due to the strong herd protection induced by PCV7 in high income countries, even at low coverage levels, which has helped to control VT transmission and as a consequence VT IPD. We find some evidence of a higher impact of vaccination in settings that introduced PCV with a catchup campaign, albeit with substantial heterogeneity of PCV impact between those settings.

The method for prediction of the impact of PCV7 builds on the idea that, because of serotype replacement, the case to carrier ratio of serotypes or serotype groups is an important determinant for the success of pneumococcal conjugate vaccines, as has been proposed earlier in similar model approaches [9,14,17,18,22,26]. A model using the inherent link of carriage and IPD has been proposed for monitoring IPD through post PCV nasopharyngeal carriage in the absence of IPD surveillance and has been shown to provide valid predictions for most of the 5 studied sites [18]. Similar models have used pre-vaccination carriage prevalence and IPD incidence to predict the impact of pneumococcal vaccines on IPD. One was validated against a single estimate of IPD impact from US ABC data [22]. We present here a minimalistic tool that utilises information on the serotype distribution of carriage and IPD before vaccination to predict disease impact. We show that the predicted IPD impact closely matches the observed impact in those nine sites and that the model is able to replicate the substantial heterogeneity of impact among sites. We find that a pre-PCV serotype specific case series of IPD (e.g. through sentinel surveillance or surveillance data where the denominator data are unclear), rather than IPD incidence as used in previous approaches, and a cross-sectional carriage study, provide sufficient data to predict the percentage reduction in IPD after maturity of the PCV programme has been reached. If supplemented by pre-PCV pneumococcal disease incidence data, the absolute impact of PCV can be estimated. This study for the first time uses the observed impact of PCV7 from multiple sites to rigorously test the predictive ability of the model while showing that differences in schedule, coverage and the use of catch-up campaigns are unlikely to have had a major contribution to the observed heterogeneity in PCV7 impact on IPD. However we cannot fully rule out that other factors, including changes in testing practices and antibiotic usage, have contributed to the heterogeneity in IPD impact among settings.

The methodology employed here bases predictions on the combined information of serotype distribution in IPD and carriage prior to vaccination. It is essential that these data come from the same or well matched populations for the prediction to be valid. While data on IPD usually result from routine surveillance of a defined population which ideally is representative of the epidemiology of IPD in the country, nasopharyngeal carriage surveys may be conducted among a group that may not be representative of the population studied for IPD. If the serotype distribution in the carriage study population is not representative of that in the disease surveillance population the resulting estimate can be misleading. Similarly, if serotypes that are hardly observed in carriage (serotype 1) or epidemic serotypes (serotypes 1 or 5) contribute substantially to the local IPD burden the performance of the model may be impaired. Here we test the predictive ability of the model using IPD surveillance data from large geographic regions paired with data on nasopharyngeal carriage from relatively small population samples of the same geographic regions. We find that in instances where carriage estimates are available from different population subgroups than that of the IPD surveillance population (see Table 1) there was only limited heterogeneity between them suggesting that those samples were representative of the carriage epidemiology in the country. However, the pneumococcal serotypes are affected by secular changes [27] which could also result in a temporal mismatch of carriage and IPD data. The impact of PCV on IPD incidence in Greece and Canada was only reported in Crete and Calgary (IRR 0.61 (0.06 to 6.50) and 0.56 (0.22 to 1.42)) for which no carriage data was available. However, carriage data were available from other regions of Greece and Canada [28-31]. If used to predict the impact of PCV7 we estimated a likely IRR of 0.29 (0.00 to 1.49) and 0.46 (0.20 to 0.78) for Greece and Canada respectively assuming full serotype replacement.

Our predictions are formed on the grounds of three major assumptions: (i) that vaccine serotypes will eventually be eliminated in the post vaccination era, (ii) that VT carriage is replaced by NVT carriage to a proportion  $\lambda = 1$  (complete replacement) and (iii) that the average pathogenicity of non-vaccine serotypes remains unchanged after vaccination. In most settings almost complete elimination of both vaccine serotype carriage and disease has been observed [9–12] and the competition of NVTs has a potential supporting role in this which could help elimination even under low coverage or high infection pressure [19,32,33]. However, evidence that vaccine serotypes are eliminated in a mature programme is still sparse in low- and middle-income settings and it is yet unclear if in settings with high transmission PCVs will still induce

sufficient herd protection to interrupt VT transmission. Hence by assuming elimination of VT carriage and disease in our model we predict the potentially vaccine preventable burden when accounting for serotype replacement. We explore the prediction for scenarios without serotype replacement and with complete serotype replacement and use the latter as the most likely scenario which we validated against the observed impact of PCV7 on childhood IPD. While full serotype replacement has been frequently reported from carriage surveys in mature pneumococcal conjugate vaccination programmes [34,35] not all studies fully support this finding [10]. The presented model provides the flexibility to explore deviating assumptions on the proportion to which NVT replace VT in carriage. The average pathogenicity of the NVT group can change following vaccination if NVT serotypes of high pathogenicity disproportionally replace compared with NVT serotypes of lower pathogenicity as was observed following PCV7 when there was a substantial rise in both carriage and disease due to serotype 19A. However, the carriage prevalence rank order of serotypes were found to be associated with the size of their capsule and therefore is thought to be generally stable [36] and to increase proportionally after vaccination [14]. Furthermore the invasive potential of serotypes has found to be a globally stable property which is also constant with time [37-39].

We excluded Norway from the analysis where unlike any other setting a 40% reduction in post vaccination NVT IPD was reported [12]. Pre-PCV7 carriage information was reported from two sites in Norway leading to a pooled estimate for the proportion of VT among carriers of 0.52 (0.22 to 0.80) [40,41]. From the serotype distribution in both carriage and IPD we predict an IRR for all-serotype IPD of 0.52 (0.31 to 1.21) in a mature programme. By contrast, Feikin et al. estimate 0.17 (0.13 to 0.23) from surveillance data (S5 Fig). However, the estimate for the observed impact is based on the extrapolation of a pre vaccination increase in IPD which runs the potential risk of overestimation of the vaccine impact [42]. Without accounting for pre vaccination trends the IRR from routine IPD surveillance is estimated at 0.46 (0.32 to 0.65), as was estimated elsewhere [12,43], and is in line with our prediction. While only in Norway a significant reduction of NVT IPD in children was estimated it may reflect the inherent limitations of an ecological design to estimate the impact of vaccination after the start of routine vaccination; its susceptibility to other factors that impact on IPD incidence. We further studied the sensitivity of the models predictive value to the delay between the start of vaccination and the impact estimate from surveillance. As our base case we presented all analysis in comparison to the observed IRR three years after the introduction of PCV7. Three years were chosen as a trade-off between allowing sufficient time for herd effects and serotype replacement to stabilise and including as many data sets in the analysis as possible. We find that the predictive ability of our model is not affected by the choice of longer post vaccination periods (S6 Fig).

As estimates of PCV7 impact from low- and middle-income countries are only recently becoming available, we could only validate our method against the use of PCV7 in high-income countries and its effects in children. The validity of our results should hold for other pneumococcal disease endpoints, including non-bacteraemic pneumonia, other age groups, in particular the elderly, and conjugate vaccine formulations of higher valency; data to validate this expectation are only recently becoming available. The importance of carriage data to estimate the dynamic effects of vaccination has been increasingly recognised [44]. However, still only limited information on nasopharyngeal carriage in older age groups, particularly in elderly is collected [24] and in many settings no carriage information is available.

The potential importance of pneumococcal carriage for supporting the licensing of pneumococcal conjugate vaccines has recently been outlined [44]. We add to this by making the case for data on pneumococcal carriage to supplement that on IPD for predicting the likely impact of PCV vaccination. We present evidence that the heterogeneity in the observed impact of PCV7 on IPD is largely due to a combination of carriage and IPD of VTs and that other factors including vaccine schedule, coverage or the use of catch up campaigns have a minor role. This method could prove useful to assess the potential impact of future conjugate vaccine formulations, aid with the impact assessments of PCVs into countries where population based surveillance of IPD is not possible, and provide an impact prediction tool to countries who have not yet introduced PCV. With a growing body of evidence on the impact of different PCV formulation and from low income countries further validation will be essential to determine the full potential of this simple model.

## **Supporting Information**

**S1 Fig. Literature review flow diagram,** (TIFF)

S2 Fig. Posterior density distributions of the meta analysis on the proportion of VT carriers.

(TIFF)

**S3 Fig. Association of observed IRR with schedule, coverage and catch up campaigns (I).** Observed IRRs stratified into common schedules (top panels), coverage levels (middle panels) and catch up campaigns (lower panels). Error bars present confidence intervals as reported by Feikin et al [12].

(TIFF)

**S4 Fig. Association of observed IRR with schedule, coverage and catch up campaigns (II).** Comparison of the marginal distributions of IRRs in settings with the same schedule (top panel), same coverage levels (middle panel) and that similarly used a catch-up campaign for introduction (lower panel).

(TIFF)

**S5 Fig. Sensitivity analysis including Norway.** Comparison of predicted and observed impact of PCV7 on IPD in children younger than 5 years assuming no serotype replacement (upper panel) or full serotype replacement (lower panel). (TIF)

**S6 Fig. Sensitivity analysis including longer time-frames.** Comparison of predicted and observed impact (3, 4, 5 years after the introduction of PCV7, from left to right) of PCV7 on IPD in children younger than 5 years assuming full serotype replacement. (TIF)

**S1 Code. Model code.** (R)

## Acknowledgments

The authors like to Daniel Feikin for insightful discussions on the observed impact of PCV7 in high income settings.

## **Author Contributions**

Conceived and designed the experiments: SF WJE. Performed the experiments: SF OLPdW. Analyzed the data: SF OLPdW KLO WJE. Wrote the paper: SF OLPdW KLO WJE.

#### References

- World Health Organization (2012) Estimated Hib and pneumococcal deaths for children under 5 years of age, 2008. Available: <u>http://www.who.int/immunization\_monitoring/burden/Pneumo\_hib\_estimates/</u> en/index.html. Accessed 27 August 2013.
- Thigpen MC, Whitney CG, Messonnier NE, Zell ER, Lynfield R, et al. (2011) Bacterial meningitis in the United States, 1998–2007. N Engl J Med 364: 2016–2025. doi: <u>10.1056/NEJMoa1005384</u> PMID: <u>21612470</u>
- Griffin MR, Zhu Y, Moore MR, Whitney CG, Grijalva CG (2013) U.S. hospitalizations for pneumonia after a decade of pneumococcal vaccination. N Engl J Med 369: 155–163. doi: <u>10.1056/</u> <u>NEJMoa1209165</u> PMID: <u>23841730</u>
- Miller E, Andrews NJ, Waight PA, Slack MP, George RC (2011) Herd immunity and serotype replacement 4 years after seven-valent pneumococcal conjugate vaccination in England and Wales: an observational cohort study. Lancet Infect Dis 11: 760–768. doi: <u>10.1016/S1473-3099(11)70090-1</u> PMID: <u>21621466</u>
- Moszynski P (2011) A pneumococcal vaccine is launched in Africa to cut child deaths. BMJ 342: d1075. doi: <u>10.1136/bmj.d1075</u> PMID: <u>21325011</u>
- Moszynski P (2011) GAVI rolls out vaccines against child killers to more countries. BMJ 343: d6217. doi: <u>10.1136/bmj.d6217</u> PMID: <u>21957215</u>
- International Vaccine Access Center—Johns Hokins Bloomberg School of Public Health (2014) PCV introduction status. Available: <u>http://www.jhsph.edu/research/centers-and-institutes/ivac/resources/</u> <u>maps.html#PCV\_Intro\_status</u>. Accessed 8 May 2014.
- Johnson HL, Deloria-Knoll M, Levine OS, Stoszek SK, Freimanis Hance L, et al. (2010) Systematic evaluation of serotypes causing invasive pneumococcal disease among children under five: the pneumococcal global serotype project. PLoS Med 7: e1000348. doi: 10.1371/journal.pmed.1000348 PMID: 20957191
- Flasche S, Hoek AJ Van, Sheasby E, Waight P, Andrews N, et al. (2011) Effect of Pneumococcal Conjugate Vaccination on Serotype-Specific Carriage and Invasive Disease in England: A Cross-Sectional Study. PLoS Med 8: e1001017. doi: 10.1371/journal.pmed.1001017 PMID: 21483718
- Spijkerman J, van Gils EJM, Veenhoven RH, Hak E, Yzerman EPF, et al. (2011) Carriage of Streptococcus pneumoniae 3 years after start of vaccination program, the Netherlands. Emerg Infect Dis 17: 584–591. doi: 10.3201/eid1704.101115 PMID: 21470445
- Huang SS, Hinrichsen VL, Stevenson AE, Rifas-Shiman SL, Kleinman K, et al. (2009) Continued impact of pneumococcal conjugate vaccine on carriage in young children. Pediatrics 124: e1–e11. doi: 10.1542/peds.2008-3099 PMID: 19564254
- Feikin DR, Kagucia EW, Loo JD, Link-Gelles R, Puhan MA, et al. (2013) Serotype-specific changes in invasive pneumococcal disease after pneumococcal conjugate vaccine introduction: a pooled analysis of multiple surveillance sites. PLoS Med 10: e1001517. doi: <u>10.1371/journal.pmed.1001517</u> PMID: <u>24086113</u>
- Weinberger DM, Malley R, Lipsitch M (2011) Serotype replacement in disease after pneumococcal vaccination. Lancet 6736: 1–12. doi: <u>10.1016/S0140-6736(10)62225-8</u>.
- Nurhonen M, Auranen K (2014) Optimal serotype compositions for Pneumococcal conjugate vaccination under serotype replacement. PLoS Comput Biol 10: e1003477. doi: <u>10.1371/journal.pcbi.1003477</u> PMID: 24550722
- Cernuschi T, Furrer E, Schwalbe N, Jones A, Berndt ER, et al. (2011) Advance market commitment for pneumococcal vaccines: putting theory into practice. Bull World Health Organ 89: 913–918. doi: <u>10.</u> <u>2471/BLT.11.087700</u> PMID: <u>22271949</u>
- UNICEF (2012) Developing a strategy to support new vaccine introduction in Middle Income Countries SAGE meeting, 6–8 November 2012. Available: <u>http://www.unicef.org/supply/files/UNICEF\_MIC\_</u> Strategy\_SAGE\_2012\_11-\_Handout\_Final.pdf. Accessed 8 May 2014.
- Weinberger DM, Harboe ZB, Flasche S, Scott JA, Lipsitch M (2011) Prediction of serotypes causing invasive pneumococcal disease in unvaccinated and vaccinated populations. Epidemiology 22: 199–207. doi: 10.1097/EDE.0b013e3182087634 PMID: 21646962
- Weinberger DM, Bruden DT, Grant LR, Lipsitch M, O'Brien KL, et al. (2013) Using pneumococcal carriage data to monitor postvaccination changes in invasive disease. Am J Epidemiol 178: 1488–1495. doi: <u>10.1093/aje/kwt156</u> PMID: <u>24013204</u>
- Lipsitch M (1997) Vaccination against colonizing bacteria with multiple serotypes. Proc Natl Acad Sci U S A 94: 6571–6576. PMID: <u>9177259</u>
- Melegaro A, Choi YH, George R, Edmunds WJ, Miller E, et al. (2010) Dynamic models of pneumococcal carriage and the impact of the Heptavalent Pneumococcal Conjugate Vaccine on invasive pneumococcal disease. BMC Infect Dis 10: 90. doi: <u>10.1186/1471-2334-10-90</u> PMID: <u>20377886</u>

- Choi YH, Jit M, Flasche S, Gay N, Miller E (2012) Mathematical modelling long-term effects of replacing Prevnar7 with Prevnar13 on invasive pneumococcal diseases in England and Wales. PLoS One 7: e39927. doi: <u>10.1371/journal.pone.0039927</u> PMID: <u>22808073</u>
- Shea KM, Weycker D, Stevenson a E, Strutton DR, Pelton SI (2011) Modeling the decline in pneumococcal acute otitis media following the introduction of pneumococcal conjugate vaccines in the US. Vaccine 29: 8042–8048. doi: 10.1016/j.vaccine.2011.08.057 PMID: 21871518
- Pilishvili T, Lexau C, Farley MM, Hadler J, Harrison LH, et al. (2010) Sustained reductions in invasive pneumococcal disease in the era of conjugate vaccine. J Infect Dis 201: 32–41. doi: <u>10.1086/648593</u> PMID: <u>19947881</u>
- Le Polain de Waroux O, Flasche S, Prieto-Merino D, Edmunds WJ (2014) Age-dependent prevalence of nasopharyngeal carriage of streptococcus pneumoniae before conjugate vaccine introduction: a prediction model based on a meta-analysis. PLoS One 9: e86136. doi: <u>10.1371/journal.pone.0086136</u> PMID: <u>24465920</u>
- 25. R Core Team (2014) R: A Language and Environment for Statistical Computing.
- Yildirim I, Hanage WP, Lipsitch M, Shea KM, Stevenson A, et al. (2010) Serotype specific invasive capacity and persistent reduction in invasive pneumococcal disease. Vaccine 29: 283–288. doi: <u>10.1016/j.vaccine.2010.10.032</u> PMID: <u>21029807</u>
- Black S (2010) The volatile nature of pneumococcal serotype epidemiology: potential for misinterpretation. Pediatr Infect Dis J 29: 301–303. doi: <u>10.1097/INF.0b013e3181c391fb</u> PMID: <u>19935444</u>
- Levidiotou S, Vrioni G, Tzanakaki G, Pappa C, Gesouli H, et al. (2006) Serotype distribution of Streptococcus pneumoniae in north-western Greece and implications for a vaccination programme. FEMS Immunol Med Microbiol 48: 179–182. doi: 10.1111/j.1574-695X.2006.00126.x PMID: 17064274
- Paraskakis I, Kafetzis DA, Chrisakis A, Papavasilliou H, Kirikou H, et al. (2006) Serotypes and antimicrobial susceptibilities of 1033 pneumococci isolated from children in Greece during 2001–2004. Clin Microbiol Infect 12: 490–493. doi: <u>10.1111/j.1469-0691.2006.01383.x</u> PMID: <u>16643530</u>
- 30. Syrogiannopoulos G a, Katopodis GD, Grivea IN, Beratis NG (2002) Antimicrobial use and serotype distribution of nasopharyngeal Streptococcus pneumoniae isolates recovered from Greek children younger than 2 years old. Clin Infect Dis 35: 1174–1182. doi: <u>10.1086/343824</u> PMID: <u>12410477</u>
- Kellner JD (1999) Streptococcus pneumoniae carriage in children attending 59 canadian child care centers. Arch Pediatr Adolesc Med 153: 495. doi: 10.1001/archpedi.153.5.495 PMID: 10323630
- 32. Flasche S, Edmunds WJ, Miller E, Goldblatt D, Robertson C, et al. (2013) The impact of specific and non-specific immunity on the ecology of Streptococcus pneumoniae and the implications for vaccination. Proc Biol Sci 280: 20131939. doi: 10.1098/rspb.2013.1939 PMID: 24089337
- Bottomley C, Roca A, Hill PC, Greenwood B, Isham V (2013) A mathematical model of serotype replacement in pneumococcal carriage following vaccination. J R Soc Interface 10: 20130786. doi: <u>10.</u> <u>1098/rsif.2013.0786</u> PMID: <u>24132203</u>
- Flasche S, Van Hoek AJ, Sheasby E, Waight P, Andrews N, et al. (2011) Effect of pneumococcal conjugate vaccination on serotype-specific carriage and invasive disease in England: a cross-sectional study. PLoS Med 8: e1001017. doi: 10.1371/journal.pmed.1001017 PMID: 21483718
- Vestrheim DF, Høiby EA, Aaberge IS, Caugant D a (2010) Impact of a pneumococcal conjugate vaccination program on carriage among children in Norway. Clin Vaccine Immunol 17: 325–334. doi: <u>10.</u> <u>1128/CVI.00435-09</u> PMID: <u>20107006</u>
- 36. Weinberger DM, Trzccinski K, Lu Y- J, Bogaert D, Brandes A, et al. (2009) Pneumococcal capsular polysaccharide structure predicts serotype prevalence. PLoS Pathog 5: e1000476. doi: <u>10.1371/journal.ppat.1000476</u> PMID: <u>19521509</u>
- Brueggemann AB, Peto TE a, Crook DW, Butler JC, Kristinsson KG, et al. (2004) Temporal and geographic stability of the serogroup-specific invasive disease potential of Streptococcus pneumoniae in children. J Infect Dis 190: 1203–1211. doi: 10.1086/423820 PMID: 15346329
- Van Hoek AJ, Sheppard CL, Andrews NJ, Waight PA, Slack MPE, et al. (2014) Pneumococcal carriage in children and adults two years after introduction of the thirteen valent pneumococcal conjugate vaccine in England. Vaccine. doi: 10.1016/j.vaccine.2014.03.017.
- Scott JR, Millar E V, Lipsitch M, Moulton LH, Weatherholtz R, et al. (2012) Impact of more than a decade of pneumococcal conjugate vaccine use on carriage and invasive potential in Native American communities. J Infect Dis 205: 280–288. doi: 10.1093/infdis/jir730 PMID: 22128315
- 40. Vestrheim DF, Høiby EA, Aaberge IS, Caugant D a (2008) Phenotypic and genotypic characterization of Streptococcus pneumoniae strains colonizing children attending day-care centers in Norway. J Clin Microbiol 46: 2508–2518. doi: <u>10.1128/JCM.02296-07</u> PMID: <u>18524970</u>

- Sogstad MKR, Aaberge IS, Sørdal JO, Høiby EA, Frøholm LO, et al. (2006) Carriage of Streptococcus pneumoniae in healthy Norwegian children attending day-care centres. Eur J Clin Microbiol Infect Dis 25: 510–514. doi: <u>10.1007/s10096-006-0177-0</u> PMID: <u>16896824</u>
- Flasche S, Slack M, Miller E (2011) Long term trends introduce a potential bias when evaluating the impact of the pneumococcal conjugate vaccination programme in England and Wales. Eurosurveillance 16: 1–6.
- 43. Steens A, Bergsaker M a R, Aaberge IS, Rønning K, Vestrheim DF (2013) Prompt effect of replacing the 7-valent pneumococcal conjugate vaccine with the 13-valent vaccine on the epidemiology of invasive pneumococcal disease in Norway. Vaccine: 1–7. doi: 10.1016/j.vaccine.2013.10.032 PMID: 24055086
- Goldblatt D, Ramakrishnan M, O'Brien K (2013) Using the impact of pneumococcal vaccines on nasopharyngeal carriage to aid licensing and vaccine implementation; a PneumoCarr meeting report March 27–28, 2012, Geneva. Vaccine 32: 146–152. doi: <u>10.1016/j.vaccine.2013.06.040</u> PMID: <u>23933374</u>
- 45. Sharma D, Baughman W, Holst A, Thomas S, Jackson D, et al. (2013) Pneumococcal carriage and invasive disease in children before introduction of the 13-valent conjugate vaccine: comparison with the era before 7-valent conjugate vaccine. Pediatr Infect Dis J 32: e45–e53. doi: <u>10.1097/INF.</u>0b013e3182788fdd PMID: 23080290
- 46. Hammitt LL, Bruden DL, Butler JC, Baggett HC, Hurlburt D a, et al. (2006) Indirect effect of conjugate vaccine on adult carriage of Streptococcus pneumoniae: an explanation of trends in invasive pneumo-coccal disease. J Infect Dis 193: 1487–1494. doi: <u>10.1086/503805</u> PMID: <u>16652275</u>
- Moore MR, Hyde TB, Hennessy TW, Parks DJ, Reasonover AL, et al. (2004) Impact of a conjugate vaccine on community-wide carriage of nonsusceptible Streptococcus pneumoniae in Alaska. J Infect Dis 190: 2031–2038. doi: <u>10.1086/425422</u> PMID: <u>15529269</u>
- Stubbs E, Hare K, Wilson C, Morris P, Leach AJ (2005) Streptococcus pneumoniae and noncapsular Haemophilus influenzae nasal carriage and hand contamination in children. Pediatr Infect Dis J 24: 423–428. doi: 10.1097/01.inf.0000160945.87356.ca PMID: 15876941
- 49. Harboe ZB, Slotved H- C, Konradsen HB, Kaltoft MS (2012) A pneumococcal carriage study in danish pre-school children before the introduction of pneumococcal conjugate vaccination. Open Microbiol J 6: 40–44. doi: 10.2174/1874285801206010040 PMID: 22611459
- Brueggemann AB, Griffiths DT, Meats E, Peto T, Crook DW, et al. (2003) Clonal relationships between invasive and carriage Streptococcus pneumoniae and serotype- and clone-specific differences in invasive disease potential. J Infect Dis 187: 1424–1432. doi: 10.1086/374624 PMID: 12717624
- Roche A, Heath PT, Sharland M, Strachan D, Breathnach A, et al. (2007) Prevalence of nasopharyngeal carriage of pneumococcus in preschool children attending day care in London. Arch Dis Child 92: 1073–1076. doi: <u>10.1136/adc.2007.126359</u> PMID: <u>17768150</u>
- Lakshman R, Murdoch C, Race G, Burkinshaw R, Shaw L, et al. (2003) Pneumococcal nasopharyngeal carriage in children following heptavalent pneumococcal conjugate vaccination in infancy. Arch Dis Child 88: 211–214. PMID: <u>12598380</u>
- Salt P, Banner C, Oh S, Yu L, Lewis S, et al. (2007) Social mixing with other children during infancy enhances antibody response to a pneumococcal conjugate vaccine in early childhood. Clin Vaccine Immunol 14: 593–599. doi: 10.1128/CVI.00344-06 PMID: 17344347
- Sleeman KL, Daniels L, Gardiner M, Griffiths D, Deeks JJ, et al. (2005) Acquisition of Streptococcus pneumoniae and nonspecific morbidity in infants and their families. Pediatr Infect Dis J 24: 121–127. doi: 10.1097/01.inf.0000151030.10159.b1 PMID: 15702039
- 55. Millar E V, O'Brien KL, Zell ER, Bronsdon MA, Reid R, et al. (2009) Nasopharyngeal carriage of Streptococcus pneumoniae in Navajo and White Mountain Apache children before the introduction of pneumococcal conjugate vaccine. Pediatr Infect Dis J 28: 711–716. doi: <u>10.1097/INF.0b013e3181a06303</u> PMID: <u>19593248</u>
- 56. Yeh SH, Zangwill KM, Lee H, Chang S- J, Wong VI, et al. (2003) Heptavalent pneumococcal vaccine conjugated to outer membrane protein of Neisseria meningitidis serogroup b and nasopharyngeal carriage of Streptococcus pneumoniae in infants. Vaccine 21: 2627–2631. doi: <u>10.1016/S0264-410X(03)</u> 00039-2 PMID: <u>12744899</u>
- 57. Van Gils EJM, Veenhoven RH, Hak E, Rodenburg GD, Bogaert D, et al. (2009) Effect of reduced-dose schedules with 7-valent pneumococcal conjugate vaccine on nasopharyngeal pneumococcal carriage in children: a randomized controlled trial. JAMA 302: 159–167. doi: <u>10.1001/jama.2009.975</u> PMID: 19584345
- Labout JAM, Duijts L, Arends LR, Jaddoe VW V, Hofman A, et al. (2008) Factors associated with pneumococcal carriage in healthy Dutch infants: the generation R study. J Pediatr 153: 771–776. doi: <u>10.1016/j.jpeds.2008.05.061</u> PMID: <u>18621390</u>